Author's response to reviews

Title: The Association of Depression and Anxiety with Glycemic Control among Mexican Americans with Diabetes Living Near the U.S.-Mexico Border

Authors:

Darla E Kendzor (Darla.Kendzor@UTSouthwestern.edu)
Minxing Chen (MChen3@mdanderson.org)
Belinda M Reininger (Belinda.M.Reininger@uth.tmc.edu)
Michael S Businelle (Michael.Businelle@UTSouthwestern.edu)
Diana W Stewart (DWStewart@mdanderson.org)
Susan P Fisher-Hoch (Susan.P.Fisher-Hoch@uth.tmc.edu)
Anne R Rentfro (Anne.Rentfro@utb.edu)
David W Wetter (dwetter@mdanderson.org)
Joseph B McCormick (Joseph.B.McCormick@uth.tmc.edu)

Version: 2 Date: 15 April 2013

Author’s response to reviews: see over